Author’s response to reviews

Title: Impact of the introduction of ultrasound services in a limited resource setting: Rural Rwanda 2008.

Authors:

Sachita P. Shah (sachita.shah@gmail.com)
Henry Epino (hepino@pih.org)
Gene Bukhman (gbukhman@gmail.com)
Irenee Umulisa (umulisa5@yahoo.fr)
JMV Dushimiyimana (dushimiyimana@gmail.com)
Andrew Reichman (reichman.andrew@gmail.com)
Vicki E. Noble (vnoble@partners.org)

Version: 3 Date: 23 March 2009

Author’s response to reviews:

Thank you for your suggestions.

We have made the following formatting changes

1. The abstract is now less than 350 words,

2. We made the required revisions to the author list so that the title page and the manuscript have the same order and detail.

3. We have edited the references to reflect the BMC style. Reference #3 is a challenge as it is a report but hopefully now it conforms to the style suggested.

4. We have added a Figure legend and have removed the titles and figure numbers from the actual figures.

5. We have separated Figure 1 and 2 into two files.

6. We have deleted the table from additional files and have uploaded it to the table section.

7. We have changed the citations from superscripts to bracketed numbers.

8. We have deleted capitals from the title.
9. We have renamed the figures using roman numerals.

I think that is it!

Thank you

Vicki Noble